Author's response to reviews

Title: Diabetic Neuropathic Cachexia: Challenges and Priorities of Clinical Practice; A case report.

Authors:

Deeb Daoud Naccache (d_deeb@rambam.health.gov.il)
William Nseir (w.nseir@yahoo.com)
Moshe Herskovitz (m_hershkovitz@rambam.health.gov.il)
Mogher Khamaisi (mogher.Khamaisi@joslin.harvard.edu)

Version: 5  Date: 12 November 2013

Author's response to reviews: see over
Your manuscript has now been peer reviewed and the comments are below.

1. Please restructure the Abstract section on page 2. The abstract must not exceed 350 words. *Abstract word count is 192 words.* Please do not use abbreviations or references in the abstract. *Abbreviations were corrected as the reviewer indicated.* The abstract should be structured into three sections and should make clear how the case report adds to the medical literature: *Abstract was restructured as the reviewer indicated.*

   > Introduction - An introduction about why this case is important and needs to be reported. Please include information on whether this is the first report of this kind in the literature.
   > Case presentation - Brief details of what the patient(s) presented with, including the patient’s age, sex and ethnic background. If the Case reported is of multiple patients, please enumerate them accordingly.
   > Conclusion - A brief conclusion of what the reader should learn from the case report and what the clinical impact will be. Is it an original case report of interest to a particular clinical specialty of medicine or will it have a broader clinical impact across medicine? Please include information on how it will significantly advance our knowledge of a particular disease etiology or drug mechanism.

2. Please include the Patient’s perspective after the Conclusions section. Please ensure that all patient identifiers are removed to maintain confidentiality. *Patient’s perspective was placed and patient identifiers were removed as the reviewer indicated.*